

Patient Journey of Generalized Pustular Psoriasis: A Real-world Study Using Data-mining Methods and Japanese Claims Data

Koremasa HAYAMA¹, Yahui TIAN^{2*}, Ryoko IWASAKI^{3*} and Hideki FUJITA¹

¹Division of Cutaneous Science, Department of Dermatology, Nihon University School of Medicine, Tokyo, Japan, ²Boehringer Ingelheim Pharmaceuticals, Inc., Ridgefield, CT, USA and ³Nippon Boehringer Ingelheim Co., Ltd, Tokyo, Japan. *E-mails: yahui.tian@boehringeringelheim.com; ryoko.iwasaki@boehringer-ingelheim.com

Submitted May 30, 2023. Accepted Dec 12, 2023;

Published XX. DOI: 10.2340/actadv.v104.11946. Acta Derm Venereol 2024; 104: adv11946.

Generalized pustular psoriasis (GPP) is a chronic, rare, autoinflammatory skin disease characterized by widespread eruption of sterile pustules with or without systemic symptoms (1, 2). Owing to the rarity of GPP, there is a lack of international consensus on diagnosis and management, and a paucity of standardized global guidelines (3). Although there are established GPP guidelines in Japan (2), disease understanding among general dermatologists is limited. Moreover, several pustular skin conditions have similar clinical presentations, which contribute to the diagnostic complexity of GPP (3).

A greater understanding of disease trajectory before GPP diagnosis may be key to improving the timeliness and accuracy of diagnosis, and subsequently, patient outcomes. However, thus far, disease trajectories in patients with GPP have not been ascertained on a population level.

Using population data from a Japanese health insurance database, this study describes the journey of patients with GPP and explore other diagnoses received before GPP. Notably, the methods used in this study (Appendix S1) may be used to mine databases from other countries.

MATERIALS AND METHODS

This retrospective cohort study used two data mining and machinelearning models with different medical foci to evaluate anonymized claims data from the Japan Medical Data Center (JMDC), from 1 January 2005 to 29 February 2020. Eligible patients with an incident diagnosis of GPP (International Classification of Diseases 10th Revision code L40.1) and \geq 1 year of continuous insurance enrolment before their index date were followed from the first confirmed diagnosis of any disease to the first confirmed GPP diagnosis (index date; Fig. S1); their journey before GPP diagnosis was analyzed. This study was approved by the ethics committee of the Nishi-Umeda Clinic for Asian Medical Collaboration, Osaka, Japan; approval number AMC-BI-21-009.

Data were analyzed using network analysis and sequential pattern mining. Baseline covariates, including age and sex, are presented descriptively.

RESULTS

Of the 435 patients in the JMDC database with GPP diagnosis, 189 met the inclusion criteria. The median age was 39 (range 1–74) years and 42.9% of patients were female (Table SI).

Overall, 31/607 diseases diagnosed before a GPP diagnosis appeared in the network pattern with a direct or indirect connection to GPP. Twenty-two diseases were diagnosed in \geq 20% of patients (**Fig. 1**; Table SII), including atopic conditions (e.g. conjunctivitis and atopic dermatitis; Table SII). Fifteen diseases were directly linked to GPP (Fig. S2); the most common being allergic rhinitis (53.5%), acute bronchitis (52.9%), and acute upper respiratory tract infection (48.1%; Fig. 1).



Fig. 1. Diagnoses (n = 22) before a generalized pustular psoriasis (GPP) diagnosis occurring in $\geq 20\%$ of patients. URT: upper respiratory tract.

Published by Medical Journals Sweden, on behalf of the Society for Publication of Acta Dermato-Venereologica. This is an Open Access article distributed under the terms of the Creative Commons Attribution-NonCommercial 4.0 International License (https://creativecommons.org/licenses/by-nc/4.0/)

ActaDV

Psoriasis vulgaris (PsV) was a pre-diagnosis in 29.4% of patients. Time to GPP diagnosis for patients with any of the 15 diseases directly linked to GPP is shown in Fig. S3 and Table SII. Sixty-seven sites were identified as making pre-GPP diagnoses of interest, with clinics (any type of clinic including but not limited to dermatology and internal general medicine) and other hospitals (any hospital that is not a clinic, public hospital, or University hospital) the most frequent settings (n=38/67 [56.7%] and n=14/67 [20.9%], respectively (Fig. S4)).

To explore the short-term patient journey regardless of frequency, we extracted disease diagnoses first made ≤ 120 days before GPP diagnosis. Pre-GPP diagnoses occurring in $\geq 1\%$ of patients included skin diseases or symptoms, and differential diagnoses; these occurred at a relatively low frequency (1.1–6.4%; Fig. S5). Table SIII shows diseases that were diagnosed ≤ 120 days before GPP diagnosis.

DISCUSSION

This real-world study was the first to explore disease trajectories and the pre-diagnosis history of patients with GPP. The study identified 22 diseases diagnosed in \geq 20% of patients before GPP diagnosis, and highlighted direct links between 15 diseases and GPP. Five of the 15 diseases linked to GPP were last diagnosed \leq 120 days before GPP diagnosis, including PsV (known to precede GPP in some patients (4)) and other diseases (e.g. impetigo) that often represent differential diagnoses for GPP. These data align with previous research from the Japanese Society for Psoriasis Research, reporting PsV prevalence of 31–78% among Japanese, Malaysian, and Portuguese patients with GPP (4–9).

The frequency of diseases that were first diagnosed \leq 120 days before GPP diagnosis was low, indicating that they were minor diagnoses in the patient journey. Overall, several skin diseases and dermatological signs/symptoms were identified (Fig. S5) that may exhibit similar symptoms to those of early GPP. For example, fever is common to both cellulitis and GPP, and pustules, skin erosion, and erythema present at initial GPP onset (2). Additionally, patients experiencing severe itch can be mistakenly diagnosed with eczema. This highlights the difficulty in correctly diagnosing early GPP symptoms and the need for dermatologists to consider differential diagnoses. A high proportion of initial diagnoses in this study were made in clinics including but not limited to dermatology clinics. Unfortunately, data on the type of clinic were unavailable for the most of these diagnoses, so the impact of dermatologist involvement on misdiagnosis remains unclear. Overall, our findings highlight the challenge of accurate GPP diagnosis, similar to the results of a US patient survey, in which 59% of patients with GPP were misdiagnosed before receiving a confirmed GPP diagnosis (4, 10).

The short-term patient journey to a GPP diagnosis may include diseases that trigger GPP. Although there is some overlap between GPP symptoms and cellulitis that may lead to misdiagnosis (e.g. swelling, redness, and fever), other symptoms differ; cellulitis has been reported as a trigger for GPP (2). Pregnancy is a known trigger for GPP and this was reflected our study, with threatened premature delivery identified ≤ 120 days before GPP diagnosis (Table SIII). Conditions identified \leq 120 days before GPP diagnosis included allergic contact dermatitis and allergic rhinoconjunctivitis, for which corticosteroids may be prescribed (11, 12). As corticosteroid withdrawal is a known trigger for GPP, the treatment, rather than the disease per se, could be the trigger. Indeed, it was difficult to establish a relationship between diseases directly connected to a GPP diagnosis with the onset of GPP, particularly for patients diagnosed several months before GPP. For example, diagnoses of myopic astigmatism and impacted cerumen occurred approximately 285 and 326 days prior to GPP diagnosis. Although there is currently no evidence for the link between these conditions and GPP, a potential correlation cannot be completely disregarded. In addition, as the most common diagnoses occurring in \geq 5% of patients during the pre-GPP diagnosis period are relatively common conditions (allergic rhinitis, acute bronchitis, and acute upper respiratory tract infection) it is not easy to decipher their relevance to GPP onset.

This study included a representative sample of patients with incident GPP diagnoses; however, external validity may be limited due to the restriction to a single Japanese database that includes only 4 million of ~125 million people in Japan (13), and few individuals aged >65 years. Our study population comprised younger patients compared with other Japanese studies (7, 14) and a higher proportion of males than females. While most GPP studies observe a higher proportion of females versus males (15), a higher proportion of males has been observed in other Japanese studies (13), including one utilizing JMDC data (16). As employed individuals comprise the majority of patients captured in the JMDC database, older, unemployed or female patients (who make up a large proportion of the Japanese population) may be under-represented. There was also a potential risk of bias due to the retrospective study design, restricting the analysis to patients with ≥ 1 year of continuous insurance enrolment, and relying on the quality and availability of data in the database, all of which could have included inconsistencies or errors. Furthermore, using the JMDC database to analyse GPP diagnoses has not yet been validated.

This first real-world study exploring the short- and long-term disease trajectories in patients before their GPP diagnosis found that approximately 30% had received a prior PsV diagnosis. Other pre-GPP diagnoses in the patient journey may represent differential diagnoses that clinicians need to exclude when diagnosing GPP or that may trigger GPP.

ACKNOWLEDGEMENTS

The study was supported and funded by Boehringer Ingelheim.

We would like to thank Dr Fei Wang of Cornell University for his expert and helpful discussions. We extend our thanks to our colleagues Christina Raabe, Vivek Jha, Sumana Nagendrakumar, Nan Shao, Zheng Zhu, Ziwei Yin, and Linwei Li for their contributions and support in this study. In the preparation of this manuscript, Isabella Goldsbrough Alves, PhD, Alexander Morrison, MSc, and Clare Bellward, MRes, of Hyperion, OPEN Health Communications, London UK, provided medical writing, editorial, and formatting support, which was contracted and funded by Boehringer Ingelheim.

Data sharing statement: To ensure independent interpretation of clinical study results and enable authors to fulfil their role and obligations under the ICMJE criteria, Boehringer Ingelheim grants all external authors access to clinical study data pertinent to the development of the publication. In adherence with the Boehringer Ingelheim Policy on Transparency and Publication of Clinical Study Data, scientific and medical researchers can request access to clinical study data when it becomes available on Vivli - Center for Global Clinical Research Data, and earliest after publication of the primary manuscript in a peer-reviewed journal, regulatory activities are complete, and other criteria are met. Please visit Medical & Clinical Trials | Clinical Research | MyStudyWindow for further information.

Conflicts of interest: The authors met criteria for authorship as recommended by the International Committee of Medical Journal Editors (ICMJE). The authors did not receive payment related to the development of this manuscript. Boehringer Ingelheim was given the opportunity to review the manuscript for medical and scientific accuracy as well as intellectual property considerations. KH declares receiving grants and/or personal fees from AbbVie, Boehringer Ingelheim, Eisai, Janssen Pharmaceuticals, Maruho, Mitsubishi Tanabe Pharma, Novartis, Sun Pharmaceutical Industries, and Taiho Pharmaceutical. YT and RI are employees of Boehringer Ingelheim. HF declares receiving honoraria or fees for serving on advisory boards, as a speaker and as a consultant, as well as grants as an investigator from AbbVie, Amgen, Boehringer Ingelheim, Bristol Myers Squibb, Daiichi Sankyo, Eisai, Eli Lilly, Janssen Pharmaceuticals, Japan Blood Products Organization, JMEC, Kaken, Kyowa Kirin, LEO Pharma, Maruho, Mitsubishi Tanabe Pharma, Nihon Pharmaceutical, Novartis, Otsuka Pharmaceutical, Sanofi, Sato Pharmaceutical, Sun Pharmaceutical Industries, Taiho Pharmaceutical, Torii Pharmaceutical, UCB, and Ushio.

REFERENCES

- Navarini AA, Burden AD, Capon F, Mrowietz U, Puig L, Köks S, et al. European consensus statement on phenotypes of pustular psoriasis. J Eur Acad Dermatol Venereol 2017; 31: 1792–1799.
- 2. Fujita H, Terui T, Hayama K, Akiyama M, Ikeda S, Mabuchi T,

et al. Japanese guidelines for the management and treatment of generalized pustular psoriasis: the new pathogenesis and treatment of GPP. J Dermatol 2018; 45: 1235–1270.

- Fujita H, Gooderham M, Romiti R. Diagnosis of generalized pustular psoriasis. Am J Clin Dermatol 2022; 23: 31–38.
- Choon SE, Lai NM, Mohammad NA, Nanu NM, Tey KE, Chew SF. Clinical profile, morbidity, and outcome of adult-onset generalized pustular psoriasis: analysis of 102 cases seen in a tertiary hospital in Johor, Malaysia. Int J Dermatol 2014; 53: 676–684.
- Ohkawara A, Yasuda H, Kobayashi H, Inaba Y, Ogawa H, Hashimoto I, et al. Generalized pustular psoriasis in Japan: two distinct groups formed by differences in symptoms and genetic background. Acta Derm Venereol 1996; 76: 68–71.
- Borges-Costa J, Silva R, Gonçalves L, Filipe P, de Almeida LS, Gomes MM. Clinical and laboratory features in acute generalized pustular psoriasis. Am J Clin Dermatol 2011; 12: 271–276.
- Ohata C, Tsuruta N, Yonekura K, Higashi Y, Saito K, Katayama E, et al. Clinical characteristics of Japanese pustular psoriasis: a multicenter observational study. J Dermatol 2022; 49: 142–150.
- Lau BW, Lim DZ, Capon F, Barker JN, Choon SE. Juvenile generalized pustular psoriasis is a chronic recalcitrant disease: an analysis of 27 patients seen in a tertiary hospital in Johor, Malaysia. Int J Dermatol 2017; 56: 392–399.
- Kamiya K, Ohtsuki M. Epidemiological survey of patients with pustular psoriasis in the Japanese Society for Psoriasis Research from 2017 to 2020. J Dermatol 2023; 50: 3–11.
- Reisner DV, Johnsson FD, Kotowsky N, Brunette S, Valdecantos, W, Eyerich K. Impact of generalized pustular psoriasis from the perspective of people living with the condition: results of an online survey. Am J Clin Dermatol 2022; 23: 65–71.
- Ciprandi G, Leonardi S, Zicari AM, Tosca MA, Marseglia G. Allergic rhinoconjunctivitis: pathophysiological mechanism and new therapeutic approach. Acta Biomed 2020; 91: 93–96.
- Kostner L, Anzengruber F, Guillod C, Recher M, Schmid-Grendelmeier P, Navarini AA. Allergic contact dermatitis. Immunol Allergy Clin North Am 2017; 37: 141–152.
- e-Stat. Population estimates by age (five-year groups) and sex – December 1, 2022 (Final estimates), May 1, 2023 (Provisional estimates). Tokyo: Statistics Bureau of Japan.
- 14. Miyachi H, Konishi T, Kumazawa R, Matsui H, Shimizu S, Fushimi K, et al. Treatments and outcomes of generalized pustular psoriasis: a cohort of 1516 patients in a nationwide inpatient database in Japan. J Am Acad Dermatol 2022; 86: 1266–1274.
- Kharawala S, Golembesky AK, Bohn RL, Esser D. The clinical, humanistic, and economic burden of generalized pustular psoriasis: a structured review. Expert Rev Clin Immunol 2020; 16: 239–252.
- Okubo Y, Kotowsky N, Gao R, Saito K, Morita A. Clinical characteristics and health-care resource utilization in patients with generalized pustular psoriasis using real-world evidence from the Japanese Medical Data Center database. J Dermatol 2021; 48: 1675–1687.